**Author's response to reviews**

**Title:** The development of socio-economic health differences in childhood: results of the Dutch longitudinal PIAMA birth cohort.

**Authors:**

Annemarie Ruijsbroek (annemarie.ruijsbroek@rivm.nl)
Alet H Wijga (alet.wijga@rivm.nl)
Marjan Kerkhof (m.kerkhof@epi.umcg.nl)
Gerard H Koppelman (g.h.koppelman@bkk.umcg.nl)
Henriette A Smit (H.A.Smit@umcutrecht.nl)
Mariël Droomers (mariel.droomers@rivm.nl)

**Version:** 4  **Date:** 21 March 2011

**Author's response to reviews:**

18 March 2011

Dear editor,

Thank you for informing us about the necessary formatting changes. All requested revisions have been made:

- The statement that all authors read and approved the final manuscript is added to ‘Authors’ contributions'
- Highlights are removed from the manuscript
- the vertical lines are removed from the tables
- both figures are cropped as described in the online figure guideline
- The manuscript is checked for typographical errors.

With kind regards,
Annemarie Ruijsbroek